Author’s response to reviews

Title: Use of saliva to monitor meningococcal vaccine responses: proposing a threshold in saliva as surrogate of protection

Authors:

Bereket Malako (berytone@gmail.com)

Benedict Asamoah (benedict_oppong.asamoah@med.lu.se)

Minyahil Boltena (minyahil.t@gmail.com)

Robel Hussen (robelhussen@rocketmail.com)

Meklit Gebre (mekumeku0220@gmail.com)

Version: 3 Date: 29 Nov 2018

Author’s response to reviews:

NUTN-D-18-00135R2

Stunting and anemia among children 6-23 months old in Damot Sore District, Southern Ethiopia

Bereket Geze Malako, Masters of Science in Nutrition; Benedict Oppong Asamoah, PhD; Minyahil Tadesse, MPH; Robel Hussen, Masters of Science in Nutrition; Meklit Tesfaye Gebre, Medical Doctor

BMC Nutrition

Dear Mr Malako,

Your manuscript "Stunting and anemia among children 6-23 months old in Damot Sore District, Southern Ethiopia" (NUTN-D-18-00135R2) has been assessed by our reviewers. Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is potentially acceptable for publication in BMC Nutrition, once you have carried out some essential revisions, shown below:
1. Please ensure that all tables are cited within the text. Any items which are not cited may be deleted by our production department upon publication.

Response 1: All the tables including the figure are cited within the text

2. Please rename the header “Materials and Methods” to “Methods.”

Response 2: The header “Material and Methods” has been renamed to “Methods”